Author's response to reviews

Title: Patients visiting the complementary medicine clinic for pain: a cross sectional study

Authors:

Roni Peleg (pelegr@bgu.ac.il)
Orly Liberman (orlylib@bgu.ac.il)
Yan Press (yanp@inter.net.il)
Pesach Shvartzman (spesah@bgu.ac.il)

Version: 3 Date: 26 April 2011

Author's response to reviews: see over
April 24, 2011

Editor
BMC Complementary and Alternative Medicine

MS -1514463964912508: "Patients visiting the complementary medicine clinic for pain: a cross sectional study"

Dear Editor:

In response to the additional revisions requested by the reviewers attached is the corrected version of our manuscript.

We would like to thank the reviewers again for taking the time and forwarding their comments. Following is a point by point reference to these revisions (additions and corrections are highlighted in the revised manuscript):

Reviewer 1:
1. The conclusion section was changed in accordance with the reviewer’s suggestion in both the abstract and the manuscript.

Reviewer 2:
1. The reviewer is correct, this was an oversight, and the word chronic was deleted.
2. We agree with the reviewer and have acknowledged this statement as a limitation of the study.
3. This was deleted.
4. The background section was corrected.
5. This paragraph was moved to the background section.
6. Prediction points were changed to predictors.
7. This was corrected.
8. We accept the reviewer’s suggestion and corrected accordingly.
We hope you find the corrected version of the manuscript acceptable for publication in the Journal of BMC Complementary and Alternative Medicine.

Please do not hesitate to contact me for any further questions or clarifications.

I look forward to your reply.

Sincerely,

Roni Peleg, MD.
Corresponding Author